Author's response to reviews

Title: The Link4Health study to evaluate the effectiveness of a combination intervention strategy for linkage to and retention in HIV care in Swaziland: protocol for a cluster randomized trial

Authors:

Margaret L McNairy (mollymcnairy@gmail.com)
Averie B Gachuhi (ab3857@columbia.edu)
Matthew R Lamb (mrl2013@columbia.edu)
Harriet Nuwagaba-Biribonwoha (hn2158@columbia.edu)
Sean Burke (sb3391@columbia.edu)
Sikhathele Mazibuko (smazibuko11@gmail.com)
Ruben Sahabo (rs2462@columbia.edu)
Neena Philip (nmp6@columbia.edu)
Peter Ehrenkranz (Peter.Ehrenkranz@gatesfoundation.org)
Velephi Okello (vjokello@gmail.com)
Wafaa M El-Sadr (wme1@columbia.edu)

Version: 2
Date: 8 July 2015

Author's response to reviews:

July 8, 2015

Dear Editors:

We have made the following changes to the manuscript:

1. Title:

The new title is: The Link4Health Study to evaluate the effectiveness of a combination intervention strategy for linkage to and retention in HIV care in Swaziland: protocol for a cluster randomized trial

2. Abstract:

We have added eligibility criteria and trial progress to abstract per CONSORT abstract guidelines.

3. CONSORT checklist:

We have completed the CONSORT checklist and listed page numbers for each section is addressed. Information on the randomization process was added to page 6.

Please let us know if you have additional questions.

Margaret McNairy